Author's response to reviews

Title: Mindfulness training for smokers via the Internet: a pilot study

Authors:

James M Davis (james.m.davis@duke.edu)
Alison R Manley (amanley@uwalumni.com)
Simon B Goldberg (simonshome@gmail.com)
Kristin A Stankevitz (kastankevitz@gmail.com)
Stevens S Smith (sss@ctri.wisc.com)

Version: 3
Date: 17 July 2014

Author's response to reviews:

Dear reviewer,

I have made the requested edits to the manuscript. Please see responses below to your requests:

We must request that you include the trial registration number in your manuscript. Since registration was retrospective in your case, we would ask you to include the date of registration after the number, so that this is clear to readers.
(Done – see first paragraph of methods section)

• Please confirm whether your study was submitted to, and approved by, your institutional ethics committee, and include a statement to this effect in your manuscript. Please also ensure that the full name of your ethics committee is included in this statement.
(Done – see first paragraph of methods section)

• Please confirm whether informed consent was obtained from all participants in your study, and include a statement to this effect in your Methods section.
(Done – see first paragraph of methods section)

• Please include details in your Methods section on the inclusion and exclusion criteria that were applied to participants in your study.
(Done – see first paragraph of methods section)

• Please ensure that your manuscript conforms to the CONSORT guidelines for the reporting of clinical trials (http://www.consort-statement.org/). Please also provide us with a completed copy of the CONSORT checklist for your study. We would suggest that you upload this as an Additional file alongside your manuscript.
(Done – see consort checklist attached.)
I hope these changes are satisfactory,

Sincerely,

James Davis